Author’s response to reviews

Title: Prevalence of hepatitis G virus infection among 67,348 blood donors in mainland China

Authors:
Taiwu Wang (wangtaiwu@hotmail.com)
Juecai Chen (cjc1025@sohu.com)
Qi Zhang (blueseal1012@126.com)
Xia Huang (xiahuangyy@163.com)
Nanzhen Xie (xienanzhen@outlook.com)
Jinhai Zhang (ahoi@163.com)
Tongjian Cai (ctjcsl@netease.com)
Yao Zhang (sydzy2003@sina.com)
Hongyan Xiong (hongyanxiong@126.com)

Version: 3 Date: 02 May 2019

Author’s response to reviews:

Dear Editor,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript again, we appreciate you very much for constructive comments and suggestions on our manuscript entitled “Prevalence of hepatitis G virus infection among 67,348 blood donors in mainland China” (ID: PUBH-D-19-00359). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We would like to express our great appreciation to you and reviewers for comments on our paper.

In this revision, we have sought for an expert for language editing. Attached please find the revised version and language editing certificate from AJE, which we would like to submit for your kind consideration. The language editing certificate was uploaded as Supplementary Material.

We sincerely hope that the revised submission is deemed suitable for publication in your journal. I look forward to receiving your decision and please do not hesitate to contact me if you have any questions.
Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Hongyan Xiong.